Author's response to reviews

Title: Disorder-specific cognitive profiles in major depressive disorder and generalized anxiety disorder.

Authors:

Sanne M Hendriks (s.hendriks@propersona.nl)
Carmilla LL Licht (c.licht@ggzingeest.nl)
Jan Spijker (j.spijker@propersona.nl)
Aartjan TF Beekman (a.beekman@ggzingeest.nl)
Florian Hardeveld (f.hardeveld@propersona.nl)
Ron de Graaf (rgraaf@trimbos.nl)
Brenda WJH Penninx (b.penninx@vumc.nl)

Version: 4
Date: 24 March 2014

Author's response to reviews: see over
March 24, 2014

Dr Alice Murray
BioMed Central
236 Gray's Inn Road
London WC1X 8HB
United Kingdom

Dear Dr. Alice Murray,

Please find enclosed a revised version of our manuscript ‘Disorder-specific cognitive profiles in major depressive disorder and generalized anxiety disorder’ (MS: 1942614638994144) for your consideration for publication in *BMC Psychiatry*.

We revised the first paragraph of the Method section and included a list of the Institutional review boards that approved our study.

‘The research protocol was approved by the Institutional review boards of the participating institutes (VU University Medical Center, University Medical Center Groningen and Leiden University Medical Center), and all respondents provided written informed consent.’

We are pleased by the positive reception and are delighted that our manuscript is, in principle, accepted for publication in *BMC Psychiatry*. We would like to thank you and the reviewers for their time and effort put in reviewing our manuscript.

Yours sincerely,

Sanne Hendriks, MD